

JGR Biogeosciences

RESEARCH ARTICLE

10.1029/2024JG008163

Key Points:

- Seagrass (*Posidonia oceanica*) meadows of the South Aegean Sea (Greece) represent important natural sinks for organic carbon, nitrogen, and carbonate.
- The accumulation pattern of organic carbon, nitrogen, and carbonate in the underlying seagrass sediments was mainly driven by sediment characteristics, followed by seagrass traits and environmental conditions.
- Fine-scale assessments are crucial for understanding the local factors shaping the storage capacity of seagrass meadows to support management decisions.

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

E. T. Apostolaki, eapost@hcmr.gr

Citation:

Apostolaki, E. T., Lavery, P. S., Litsi-Mizan, V., Serrano, E., Inostroza, K., Gerakaris, V., et al. (2024). Patterns of carbon and nitrogen accumulation in seagrass (*Posidonia oceanica*) meadows of the Eastern Mediterranean Sea. *Journal of Geophysical Research: Biogeosciences*, *129*, e2024JG008163. https://doi.org/10. 1029/2024JG008163

Received 29 MAR 2024 Accepted 27 NOV 2024

Author Contributions:

Conceptualization: Eugenia T. Apostolaki, Paul S. Lavery, Miguel A. Mateo, Oscar Serrano Formal analysis: Eugenia T. Apostolaki, Victoria Litsi-Mizan Funding acquisition: Oscar Serrano Investigation: Eugenia T. Apostolaki, Paul S. Lavery, Eduard Serrano, Karina Inostroza, Thanos Dailianis, Julius Glampedakis, Miguel A. Mateo, Oscar Serrano

© 2024. The Author(s).

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Patterns of Carbon and Nitrogen Accumulation in Seagrass (*Posidonia oceanica*) Meadows of the Eastern Mediterranean Sea

Eugenia T. Apostolaki¹, Paul S. Lavery^{2,3}, Kictoria Litsi-Mizan¹, Eduard Serrano^{3,4}, Karina Inostroza⁴, Vasilis Gerakaris⁵, Thanos Dailianis⁶, Julius Glampedakis⁶, Tara Holitzki⁷, Erik Johnson⁷, Miguel A. Mateo^{2,3}, and Oscar Serrano^{2,3}

¹Institute of Oceanography, Hellenic Centre for Marine Research, Heraklion, Greece, ²School of Science & Centre for Marine Ecosystems Research, Edith Cowan University, Joondalup, WA, Australia, ³Centre of Advanced Studies of Blanes (CEAB-CSIC), Blanes, Spain, ⁴BIOSFERA Environmental Education, Research & Conservation, L'Hospitalet de Llobregat, Spain, ⁵Institute of Oceanography, Hellenic Centre for Marine Research, Anavyssos, Greece, ⁶Institute of Marine Biology, Biotechnology and Aquaculture, Hellenic Centre for Marine Research, Heraklion, Greece, ⁷University of Hawai'i at Hilo Analytical Laboratory, Hilo, HI, USA

Abstract The variability in stocks and accumulation rates of organic carbon (C_{org}), nitrogen (N), and carbonate (CaCO₃) was studied in fifteen *Posidonia oceanica* meadows spread throughout the South Aegean Sea (Greece). In addition, the abiotic and biotic drivers determining the pattern of variability in the accumulation rates were assessed by exploring the influence of sediment characteristics, seagrass traits, and environmental settings. The meadows supported on average (±STDEV) 14.6 ± 5.0 kg C_{org} m⁻², 0.47 ± 0.17 kg N m⁻², and 249 ± 210 kg CaCO₃ m⁻² in the top meter of their sediments, with mean accumulation rates over the last 500 years of 33.6 ± 23.6 g C_{org} m⁻² yr⁻¹, 1.00 ± 0.62 g N m⁻² yr⁻¹, and 405 ± 336 g CaCO₃ m⁻² yr⁻¹ across sites. A redundancy analysis (RDA) explained 70% of the variation in C_{org}, N, and CaCO₃ accumulation rates, with three sediment characteristics (i.e., sediment C_{org}:N and C_{org}:C_{inorg} ratios and *P. oceanica* contribution to the sediment C_{org} pool) emerging as the primary set of factors shaping the accumulation of matter, followed by seagrass traits (i.e., leaf biomass and rhizome elongation) and environmental variables (i.e., suspended organic matter). The high degree of variability within the region emphasizes the need for fine-scale assessments to understand the local conditions influencing sequestration. Our findings underscored the critical role of seagrass meadows in carbon and nitrogen sequestration in the region, urging conservation efforts to protect these ecosystems and prevent potential losses of stored carbon and nitrogen following seagrass degradation.

Plain Language Summary In the present study, we explored the abiotic and biotic factors influencing the accumulation patterns of carbon, nitrogen, and carbonate in 15 seagrass (*Posidonia oceanica*) meadows across the South Aegean Sea (Greece). Sediment characteristics (i.e., sediment C_{org} :N and C_{org} : C_{inorg} ratios and *P. ocenica* contribution to the sediment C_{org} pool) mainly drove the pattern of carbon, nitrogen, and carbonate accumulation, with seagrass traits (i.e., leaf biomass and rhizome elongation) and environmental conditions (i.e., suspended organic matter) contributing to a lesser extent. Overall, seagrass meadows play a crucial role in storing carbon and nitrogen in the studied region, serving as important natural sinks, and thus, protecting these ecosystems is important to prevent the loss of these stored elements. Additionally, understanding the factors that shape the variability in *Posidonia oceanica* storage capacity is crucial for management and conservation efforts.

1. Introduction

The significant role of seagrass ecosystems in storing elements over millennia, namely carbon and nitrogen, has been documented several decades ago (Mateo et al., 1997; Romero et al., 1994; Smith, 1981). However, it is within the last decade that a plethora of information has accumulated (Costa et al., 2022) to highlight how that role could contribute toward combating climate change (Macreadie et al., 2021). Seagrass meadows have been shown to store 115.3–829.2 Mg ha⁻¹ of organic carbon (C_{org}) (Fourqurean, Kendrick, et al., 2012; Fourqurean, Duarte, et al., 2012) and 3–1,660 Mg ha⁻¹ of inorganic carbon (C_{inorg}) (Mazarrasa et al., 2015) within 1-m thick deposits, and as such, they rank among the most effective natural carbon sinks on Earth. Lately, the significant role of seagrass ecosystems as nitrogen (N) filters has emerged (Aoki et al., 2020), highlighting their potential role in



21698961

, 2024, 12, Downloa

Resources: Tara Holitzki, Erik Johnson, Oscar Serrano Visualization: Victoria Litsi-Mizan, Vasilis Gerakaris Writing – original draft: Eugenia T. Apostolaki Writing – review & editing: Paul S. Lavery, Victoria Litsi-Mizan, Eduard Serrano, Karina Inostroza, Vasilis Gerakaris, Tara Holitzki, Miguel A. Mateo, Oscar Serrano

mitigating coastal eutrophication (McGlathery et al., 2007, 2012; Reynolds et al., 2016). Seagrass ecosystems owe their high biogeochemical storage capacity to their intense metabolism (Duarte et al., 2010) and their ability to trap allochthonous organic and inorganic particles from adjacent ecosystems including terrestrial ones (Gacia et al., 2002), and to the preservation of matter in their underlying sediments (Duarte, Kennedy, et al., 2013) which is favored by the subsurface anoxic conditions and, especially, the recalcitrant nature of seagrass detritus (Kaal et al., 2019).

The size of the sink of each element exhibits high variability across seagrass habitats and species. The cycling of matter in seagrass ecosystems involves an endless fixation and loss of elements through complex and cooccurring biogeochemical processes, which result in high spatial and temporal heterogeneity in carbon and nitrogen pools in seagrass sediments. A series of processes (e.g., metabolism, mineralization, calcification, carbonate dissolution, denitrification) operate between the water and sediment interface (Barrón et al., 2014; Garcias-Bonet et al., 2018), mediated by the biota (i.e., the seagrass itself and the associated flora, fauna, and microbes (Evre et al., 2016; Trevathan-Tackett et al., 2019), and influenced by the geochemical and environmental context (Spivak et al., 2019). Some of these processes are highly interrelated; for instance, the CO_2 release from the metabolism of the community or the mineralization of organic matter could fuel carbonate (CaCO₃) dissolution in the upper sediment layer, whereas high production of CaCO₃ could increase C_{org} preservation (Macreadie et al., 2017). Furthermore, the contribution of each set of factors regulating those processes may differ depending on the element under study; variability in environmental settings, such as water column nutrients, for example, should have a greater influence on the C_{org} and N (Kindeberg et al., 2018; Lima et al., 2020) than the CaCO₃ accumulation, as both Corg and N represent more labile pools that often show a parallel pattern in accumulation (Apostolaki et al., 2022; Fourqurean, Kendrick, et al., 2012; Fourqurean, Duarte, et al., 2012; Leiva-Dueñas et al., 2023), than the CaCO₃ that has larger permanence and may also include inputs of lithogenic provenance (Saderne et al., 2019). Therefore, the magnitude of each sink depends on multiple abiotic (i.e., sediment characteristics and environmental settings) and biotic (i.e., seagrass traits) factors acting in concert at several spatiotemporal scales, and Corg, N and CaCO3 assessments, when conducted in parallel, could enhance our understanding of the variability in the capacity of seagrass meadows for elemental sequestration.

However, previous studies have explored the main factors driving this variability separately for each element, rather than in parallel. Available information shows that storage for each element varies with environmental settings (e.g., water depth, hydrodynamics), sediment characteristics (e.g., grain size, OM availability, biochemical composition and origin), and seagrass traits (e.g., production, structure), but the amount of relevant information is largely biased toward C_{org} (e.g., Kennedy et al., 2022; Lavery et al., 2013; Lima et al., 2020; Mazarrasa et al., 2021; Novak et al., 2020; Ricart et al., 2020; Röhr et al., 2016; Samper-Villarreal et al., 2016; Serrano, Lavery, López-Merino, et al., 2016) compared to N (Kindeberg et al., 2018; Leiva-Dueñas et al., 2023; Prystay et al., 2023) or CaCO₃ (Gullström et al., 2018). In addition, the available studies have investigated the factors determining the accumulation along a certain thickness of sediment (usually the top 1 m), and not during a given time period. The sediment supply dynamics and local environmental factors resulting in deposits that cover a period from several years to centuries or even millennia in some species (Monnier et al., 2021). Therefore, in such assessments, it is also important to consider the period of accumulation and not only the sediment thickness to improve our understanding of the timescales of elemental sequestration.

The iconic seagrass meadows of the Mediterranean, *Posidonia oceanica*, rank among the most important natural carbon sinks across the terrestrial and coastal realm (Röhr et al., 2018). However, available estimates show remarkable variability in stocks and accumulation rates of C_{org} , N, and CaCO₃ in the western Mediterranean (Apostolaki et al., 2022; Mazarrasa et al., 2017a; Monnier et al., 2022; Serrano, Lavery, López-Merino, et al., 2016) compared to the Eastern basin (Apostolaki et al., 2019). Although recent studies have shown that *P. oceanica* traits have a crucial contribution in shaping the significant capacity of the species to store C_{org} compared to other seagrass species (Kennedy et al., 2022), we still lack a robust explanation of the seagrass environmental and geochemical factors that determine the high C_{org} storing capacity of the species at local scales. Furthermore, a limited understanding exists regarding the factors driving the accumulation of N and CaCO₃. Therefore, fine-scale assessments are important to understand the patterns of variability across *P. oceanica* meadows and eventually allow us to identify hotspots for the implementation of blue carbon projects toward climate change mitigation.



Here, we studied the variability of carbon and nitrogen sequestration in *P. oceanica* meadows of the eastern Mediterranean Sea. We estimated the stocks and accumulation rates of C_{org} , N, and $CaCO_3$ in 15 meadows expanding along the South Aegean Sea (Greece). In addition, we assessed the abiotic and biotic drivers that determine the pattern of variability among the accumulation of C_{org} , N, and $CaCO_3$ by exploring the influence of sediment characteristics, seagrass traits, and environmental settings.

2. Materials and Methods

2.1. Study Area and Sampling

The study was conducted in the South Aegean Sea. The study included 15 monospecific *P. oceanica* meadows extending over sedimentary, metamorphic, and volcanic sediments (IGME, 1983) along approx. 3,500 km of coastline (Figure 1). The area is characterized by steep rocky cliffs, sandy shores, low precipitation, and the absence of major riverine inflow.

Within 100 m² of each meadow, the number of *P. oceanica* shoots was counted in triplicates using a 50×50 cm quadrat randomly tossed over the meadow. All shoots present within each quadrat were collected by hand, and leaves were separated and weighed. Fifteen *P. oceanica* shoots were randomly sampled from the site Gouves to calculate a mean dry weight: fresh weight ratio (DW:FW) and allow the estimation of leaf DW biomass in all 15 sites. We assumed that water content in seagrass leaves did not vary significantly among the study sites. Therefore, we used the mean DW:FW value and the fresh weight of leaves per site to estimate leaf biomass at each site. Ten orthotropic shoots were also collected at each site to allow reconstruction of rhizome elongation. The presence of seagrass species other than *P. oceanica* (e.g., *Cymodocea nodosa*) or macroalgae (e.g., *Caulerpa prolifera*) was recorded by visual census.

At each site, triplicate cores were randomly collected using manual percussion and rotation. One replicate core was sampled using a PVC pipe (2.2 m long and 7.5 cm inner diameter), while the other two replicate cores were sampled using aluminum pipes (1.5-m long and 8-cm inner diameter) with 3 cm-wide holes predrilled at 3-cm intervals along the pipe. The aluminum pipes were predrilled to effectively subsample to sediments inside the pipes in the field, a method commonly used in blue carbon science (e.g., Fourqurean, Kendrick, et al., 2012; Fourqurean, Duarte, et al., 2012). The degree of compaction was similar between PVC and aluminum pipes ($23 \pm 10\%$ and $29 \pm 8\%$, respectively). The porewater content does not have any effect on the dry bulk density calculations run to estimate carbon stocks. Potential C_{org} remineralization was not relevant considering the short period between sampling and processing of the cores (i.e., a few hours for the aluminum cores and a few days for the PVC cores) and the age of the carbon samples (i.e., from years to centuries).

The cores were inserted down to 1 m of sediment or deeper (up to 202 cm) (Table S1). The inner, outer, and total length of each core were measured after coring to allow the estimation of sediment compression, which ranged between 1.6% and 39.6% (mean \pm SD = 24 \pm 9% Table S1).

2.2. Laboratory Analysis

Fresh leaves from shoots collected at the Gouves site were weighed, dried at 60°C for 48–72 hr, and weighed again, yielding a DW:FW ratio of 0.24. Rhizome elongation was reconstructed using the lepidochronology analysis, based on the cyclic variation in sheath thickness along the rhizome (Pergent-Martini & Pergent, 1994). The thickness of each sheath per shoot was measured using a caliper. The period between two consecutive minima in sheath thickness corresponded to one lepidochronological year and, by convention, to one calendar year. The length of each rhizome segment per shoot and year was measured using a caliper.

The PVC cores were cut lengthwise, and the sediments were sliced into 1 cm-thick intervals. A subsample of sediment was collected from each hole of the aluminum cores using a modified 60-mL syringe with the end cut off. All samples were dried at 70°C until constant weight and then divided by quartering for subsequent analyses. Bulk subsamples were used for the determination of grain size, after digestion of organic matter with hydrogen peroxide (30%). Samples were sieved using 1-mm sieves, and the fractions >1 mm and <1 mm were weighed. The fraction <1 mm was analyzed using a laser diffraction particle analyzer (Mastersizer 2,000; Malvern Instruments Ltd., UK). The grain size fractions were classified as very coarse sand (>1 mm), coarse sand (0.5–1 mm), medium sand (0.25–0.5 mm), fine sand (0.063–0.25 mm), and silt/clay (<0.063 mm). Other subsamples of sediment were milled to fine powder and were used for the determination of organic carbon (C_{org}), nitrogen



Journal of Geophysical Research: Biogeosciences



Figure 1. Map of the study area in the South Aegean Sea and sites sampled (a–k). The extent of *P. oceanica* meadows at each site was obtained from Panayotidis et al. (2022). The lithology of the catchments was obtained from IGME (1983).

(N), and carbonate (CaCO₃) contents, and the stable organic carbon isotopic composition (δ^{13} C) of sediment. The determination of C_{org}, δ^{13} C, and N was done using a Costech 4010 elemental analyzer coupled to a Thermo Delta V Advantage isotope ratio mass spectrometer (UH Hilo Analytical Laboratory) at the University of Hawaii at Hilo. Prior to analysis, 50-mg powder samples were acidified in silver capsules with 4% hydrochloric acid (HCl) to remove inorganic carbon (Kennedy et al., 2005). Isotope ratios were expressed as δ values in parts per thousand (‰) relative to the Vienna Peedee Belemnite standard (δ^{13} C), using US Geological Survey standards 40 and 41. Replicate assays and standards (NIST 8704) indicated measurement errors of 0.01% for C_{org} and N contents, and 0.2‰ for δ^{13} C. The CaCO₃ content was determined using a calcimeter (Pressure Gauge Model 432 (Fann®); ASTM D 4373-84 Standard) by adding 10% HCl to the powder sample in a sealed reaction cell. The pressure built up due to the CO₂ was measured with a bourdon tube pressure gauge that was precalibrated with reagent-grade calcium carbonate.

Three to five samples of *P. oceanica* sheaths from one replicate core per site were radiocarbon-dated at the Direct AMS Radiocarbon Business Unit (Accium BioSciences, USA) following standard procedures (Stuiver & Polach, 1977) (Table S2).

2.3. Calculations and Data Analysis

Shoot density (shoots m⁻²) was estimated as the number of shoots per unit area at each site. Leaf biomass (g DW m⁻²) at each site was estimated as the product of fresh weight of leaves at the given site and the mean DW:FW ratio per unit area. Rhizome elongation (mm shoot⁻¹ yr⁻¹) at each site was estimated as the mean of annual elongation during the reconstructed period for the given site. The temporal period of reconstruction corresponded to years 1995–2016 across sites, with a maximum year span of 22 years (at Gouves) and a mean shoot age of 9.9 ± 3 years. Rhizome segments corresponding to the two most recent lepidochronological years of each shoot were excluded from the analysis, as the lepidochronological cycles may have not been complete at the time of sampling (Peirano, 2002).

The dry bulk density (DBD, g cm⁻³) of each slice was estimated by dividing the DW of sediment by the volume of the corresponding wet sample. The elemental content (% DW) was measured in alternate slices, whereas in the remaining slices, the contents were estimated as the mean content of the slices immediately above and below the slice in question. The mass of C_{org} , CaCO₃, and N (g element cm⁻²) at each slice was estimated by multiplying the DBD with the corresponding elemental content and the decompressed sediment slice thickness (in cm).

Stocks of C_{org} , N, and $CaCO_3$ (kg element m⁻²) were estimated as the cumulative mass of the corresponding element and were standardized to 1 m of sediment thickness. In cases where the length of sediment sampled was less than 1 m (in 19 out of 45 cores, Table S1), we fitted a linear regression to the downcore trends in cumulative stocks to estimate the 1-m stocks. To validate this approach, data from the cores with at least 1-m depth were used to estimate the correlation between extrapolated stocks to 1-m thickness and measured stocks in 1-m cores (Table S6), with significant correlations for C_{org} stock ($R^2 = 0.85$ and P < 0.001), N stock ($R^2 = 0.94$ and P < 0.001), and CaCO₃ stock ($R^2 = 0.97$ and P < 0.001) (Figure S1 in Supporting Information S1).

The radiocarbon ages (yr BP; Table S2) were used to produce age-depth models using the R routine "Bacon" (Blaauw & Christen, 2011), after calibration using the Marine13 Radiocarbon Age calibration curve, considering a local delta R of 110 ± 38 years (Reimer & McCormac, 2002), and assuming that the top of the core corresponded to the year of sampling (Table S1). Age-depth models for cores that only had one ¹⁴C dating were obtained by fitting a linear regression between the calibrated ¹⁴C age and the top of the core assuming that it corresponded to the year of sampling. In four cores, the ¹⁴C ages obtained were too modern to be calibrated, and therefore, we assumed that the first ¹⁴C-dated sample corresponded to 1,800 cal AD and fitted a linear regression assuming that the top of the core corresponded to the year of sampling to obtain approximate accumulation rates. This assumption is conservative and relies on the raw radiocarbon ages obtained in these four cores (ranging from 261 to 375 years BP; Table S2), which correspond to about half of the radiocarbon ages adjusted for Delta R that can be calibrated using postbomb curves (i.e., calibrated 700 years BP would render about 1,700-1,900 cal AD at our study sites (Stuiver & Reimer, 1993)). All ¹⁴ C ages reported are expressed as radiocarbon-calibrated years (cal yr BP; before present corresponding to 2018). The sediment accretion rate (SAR, cm yr⁻¹) per core was estimated from the ¹⁴C age-depth model and was standardized for the last 500 years whenever possible. In cases where the age of the core was less than 500 years (Table S1), the SAR was standardized to the oldest age possible. The SARs of the two replicate cores per site that were not radiocarbon-dated were assumed to equal the SAR in



the replicate core dated. Accumulation rates of C_{org} , N, and $CaCO_3$ (g element m⁻² yr⁻¹) of each core were derived by multiplying the corresponding cumulative mass of each element accumulated over 500 years by the corresponding SAR of the core.

2.4. Statistical Analysis

Statistical analyses were done using the accumulation rates rather than the stocks to ensure meaningful comparisons among sites. The top meter of sediment thickness, which is commonly used to assess the blue carbon storage capacity across sites, corresponded to very diverse periods of accumulation across sites, ranging from 238 years at Serifos to 2,152 years at Anavissos (Table S1). Nevertheless, we report data on C_{org} , N, and CaCO₃ stocks at the top meter (Table S1) to facilitate comparisons among other studies.

One-way analysis of variance (ANOVA) was used to test possible significant differences in accumulation rates among sites. Normality and homogeneity of variance were checked before analysis using the Shapiro-Wilk and Levene's tests, respectively. When these assumptions were not met, data were \log_{10} -transformed. If significant differences occurred (P < 0.05), Tukey's post hoc test was used to identify which sites differed.

A multivariate analysis was used to explore the abiotic and biotic factors influencing the accumulation rates across the South Aegean Sea. In particular, a redundancy analysis (RDA) was used, which performs multiple linear regression (MLR) of response variables on explanatory variables and then conducts a principal component analysis (PCA) to the matrix of fitted values of all response variables generated through the MLR. Therefore, the RDA produces two ordinations in space, one defined by the matrix of response variables and another defined by the matrix of explanatory variables. The analysis provides the total variance of the data set, partitioned into constrained variance, thereby showing how much variation in the response variables were the C_{org} , N, and CaCO₃ accumulation rates, and the explanatory variables were composed of three sets of variables that described the sediment characteristics, seagrass traits, and environmental settings.

Sediment characteristics included the mean values of grain size fractions, Corg, CaCO3, and N contents, the contribution of P. oceanica detritus to the sediment Corg pool during the last 500 years, and the lithology of the sites. We did not use Corgs N, and CaCO3 contents in the RDA to avoid any potential numerical influence they might have had on the response matrix due to their interference in the mathematical calculation of the Core, N, and CaCO3 accumulation rates, respectively. Instead, we used the Corg:N and Corg:Cinorg ratios, considering a mass ratio of 12% Cinorg in CaCO3. For determining the contribution of P. oceanica in the Corg pool, stable isotope mixing models were run for each core in R ("simmr" and "rjags" packages; (Parnell et al., 2013)). The δ^{13} C of each slice along the 500-year profile from each core was used. We acknowledge that the δ^{13} C values of P. oceanica detritus in the deeper sediment layers may have been altered during decomposition. However, very few studies have determined the δ^{13} C fractionation of seagrass species during decomposition, and their results show considerable variability, with fractionation ranging from negligible to changes up to 4% (Kelleway et al., 2022). In addition, to the best of our knowledge, none of the studies have focused on *P. oceanica*. Lastly, available decomposition studies assess the change of δ^{13} C in shorter timescales (months to a few years), while blue carbon studies refer to centennial scales. Therefore, no fractionation correlation factor between the $\delta^{13}C$ and the timeframe of our study was available. Lastly, for the multivariate statistical analysis, we used the mean values of the 500-year profile for all the geochemical variables, to integrate the variability along the profile. For the sake of comparison, using the δ^{13} C signatures along the profile rather than the ones of the superficial sediment was chosen. Potential end-members considered were as follows: "P. oceanica leaves," "Other macrophytes," and "seston and terrestrial material." "Other macrophytes" included seagrass species other than P. oceanica (i.e., C. nodosa and H. stipulacea) and macroalgae. Other seagrass species had limited presence, and macroalgae had low coverage, so they were pooled together in one group by calculating the mean value of their δ^{13} C signatures (Table S3 in Supporting Information S1). Mean δ^{13} C signatures for each source were obtained from our own unpublished data or the literature (Table S3 in Supporting Information S1). Possible differences in δ^{13} C among end-members were assessed through t-tests. Other macrophytes and seston had similar (P > 0.05) δ^{13} C signatures and were grouped to decrease uncertainty in the model outcome. The simmr_mcmc() function was used to generate the potential contribution of different sources by applying a Markov chain Monte Carlo (MCMC) (Parnell et al., 2010, 2013). In total of 10,000 iterations were conducted, from which 1,000 burn-initial iterations were discarded. The remaining samples were used to construct the posterior distribution. The convergence of each model for each site was assessed through model diagnostics, which indicated that R-hat values in all cases were close to 1, suggesting effective convergence (Vehtari et al., 2019). Finally, the lithology of each site was included as a nominal binary variable of karstic/nonkarstic mineral matrix, which served as an indication of the biogenic or lithogenic origin, respectively, of CaCO₃ in the sediments.

The seagrass variables considered were the mean shoot density, leaf biomass, and rhizome elongation measured at each site. The environmental variables taken into consideration were water depth, the concentration of chlorophyll *a* (Chl *a*) in the water column as a proxy for the nutrient regime, Secchi depth as a proxy of water transparency, and suspended particulate matter (SPM) as a proxy for turbidity. Data of Chl *a*, Secchi depth, and SPM were satellite data downloaded from the Copernicus Marine database (https://marine.copernicus.eu/). We used L4 optimally interpolated data at 4×4 km resolution (OCEANCOLOUR GLO BGC L4 MY 009 104). The "nearest neighbor" with available satellite data for each site was selected in MATLAB v. R2018a software. The distance between our sites and their "nearest neighbor" was on average 2.66 ± 1.73 km. Only in the cases of Amorgos and Milos, did the available "nearest neighbors" extend beyond the 4×4 -km grid, but they were at a maximum distance of 6 km from the sampling sites. We retrieved all available data for each variable (on diel resolution in case of Secchi depth and SPM) from the earliest measurement available (i.e., 1997) until the year of sampling. The mean value of each variable across the whole timescale available per site was used for the analysis. We acknowledge that environmental and seagrass variables referred to a shorter temporal scale than that covered by sediment variables. However, these variables were used as indicative of the environmental and seagrass context, assuming that they are representative of the period of accumulation.

Before running the RDA, possible multicollinearity among explanatory variables was assessed. First, a PCA was performed on the correlation matrix of the grain size fractions, after data were transformed using centered log ratio (Aitchison, 1986). The principal component 1 (PC1) explained 94% of the variability in grain size distribution among sites, and therefore, the PC1 scores were used in further statistical analysis (Figure S2 in Supporting Information S1). Then, a Pearson correlation was performed, and highly correlated variables (Pearson's |r| > 0.7) were removed (i.e., Chl *a*). The final data set included the following explanatory variables: PC1 scores of grain size fractions, C_{org} :N and C_{org} :C_{inorg} ratios, *P. oceanica* contribution to sediment C_{org} , lithology, shoot density, leaf biomass, rhizome elongation, water depth, Secchi depth, and SPM. The explanatory variables were examined for normality, using the Shapiro-Wilk test, and transformed with log_{10} or square root to achieve normality, if necessary. All variables were subsequently standardized using z-scores.

An initial RDA model was run, and its significance was assessed by Monte Carlo permutation test (999 permutations). Then, the most parsimonious RDA model was selected by identifying the explanatory variables that most significantly (P < 0.05) contributed to the variance of the response matrix through the Monte Carlo permutation test (backward selection, 999 permutations). A final RDA was run with those explanatory variables and tested by Monte Carlo permutation test (999 permutations) for its significance. A correlation plot (Type II scaling) was used to show the relation among the three response variables and between each response variable and each of the backward selected explanatory variables. In such plots, the relation between each pair of variables is equal to the cosine of the angle between them (i.e., a right angle describing no correlation as $\cos(90^\circ) = 0$, an acute angle, a positive correlation, and an obtuse angle, a negative correlation). Unique and shared variations among sets of variables were identified through variation partitioning (based on R^2 adjusted) and visualized in a Venn diagram.

All analyses were performed in R version 4.0.5 (R Core Team, 2021), using the vegan package (Oksanen et al., 2022).

3. Results

3.1. Variability of Corg, N, and CaCO3 Stocks and Accumulation Rates

 C_{org} stock at the top meter of sediment ranged from 7.7 ± 0.4 kg m⁻² at Metopi to 23.2 ± 7.8 kg m⁻² at Santorini 2, with a mean ± STDEV of 14.6 ± 5.0 kg m⁻² across all sites (Table S1). The range of N stock was 0.29 ± 0.09– 0.81 ± 0.02 kg m⁻² among sites, and the mean was 0.47 ± 0.17 kg m⁻². The variability of CaCO₃ stocks between sites was high, ranging from 16.6 ± 8.7 kg m⁻² at Santorini 2 to 663 ± 55 kg m⁻² at Koufonisi, having a mean value of 249 ± 210 kg m⁻².

Sediment accretion rates in the last 500 years varied from 0.05 to 0.41 cm yr⁻¹, with a mean of 0.22 ± 0.01 cm yr⁻¹ among the sites studied (Table S1). C_{org} accumulation rates over the last 500 years ranged



Journal of Geophysical Research: Biogeosciences



Figure 2. Boxplots of (a) organic carbon (C_{org}), (b) nitrogen (N), and (c) carbonate (CaCO₃) accumulation rates over the last 500 years at each site. Each box encompasses the 25% and 75% quantiles, and the horizontal dash indicates the median. Boxplots are sorted by the mean value from the lowest to the highest value of the C_{org} accumulation rate.

from 5.4 \pm 0.8 g C_{org} m⁻² yr⁻¹ at Anavissos to 78.4 \pm 22.0 g C_{org} m⁻² yr⁻¹ at Santorini 2, with a mean of 33.6 \pm 23.6 g C_{org} m⁻² yr⁻¹ across all sites (Figure 2a). Significantly higher C_{org} accumulation rates were measured at Methana, Ios, Santorini, Gouves, Kithnos, Serifos and Santorini 2 than at the remaining sites (Table S5). The N accumulation rate ranged between 0.19 \pm 0.04 at g N m⁻² yr⁻¹ at Anavissos and 2.29 \pm 0.35 g N m⁻² yr⁻¹ at Kithnos, with a mean across all sites of 1.00 \pm 0.62 g N m⁻² yr⁻¹ (Figure 2b). Overall, the N accumulation rate was higher at the sites having had also the highest C_{org} accumulation rates (i.e., Methana, Ios, Santorini, Gouves, Kithnos, Serifos, and Santorini 2), although the difference of N accumulation rates at those sites was not always significant from the rest of the sites. The accumulation rates of CaCO₃ ranged from 61 \pm 21 g CaCO₃ m⁻² yr⁻¹ at Santorini to 1,115 \pm 115 g CaCO₃ m⁻² yr⁻¹ at Metopi, with a mean value of 405 \pm 336 g CaCO₃ m⁻² yr⁻¹ across all sites and significantly higher rates at Metopi, Keros, Koufonisi, Agia Marina, and Gouves than those measured at the sites with high C_{org} accumulation rates (Figure 2c). Anavissos and Sounion had significantly lower accumulation rates than the rest of the sites for all studied elements.

3.2. Drivers of Variability in Corg, N, and CaCO3 Accumulation Rates

The temporal distribution of sediment variables in the last 500 years showed low variability at all sites (Table S1). Sediments were mainly sandy, with a high percentage of very coarse (mean \pm SE in the last 500 years across sites of 10.3 \pm 0.5%), coarse (5.9 \pm 0.4%), medium (18.4 \pm 0.5%), and fine (47.4 \pm 0.6%) sands, and silt/clay (18.0 \pm 0.7%) contents (Figure 3). Mean DBD in the last 500 years ranged between 0.52 \pm 0.01 and 1.06 \pm 0.02 g cm⁻³ across sites, with an average of 0.86 \pm 0.01 g cm⁻³ for our study area (Figure 4a). The change of C_{org} and N contents with time followed similar patterns at each site (Table S1). The mean C_{org} content in the last 500 years varied from 1.08 \pm 0.15 to 4.35 \pm 0.76% among sites, and it was on average 2.24 \pm 0.09% (Figure 4b). The range in the mean N content in the last 500 years among sites was 0.036 \pm 0.006–0.138 \pm 0.004%, and the mean for the study was 0.069 \pm 0.002% (Figure 4c). The mean CaCO₃ content in the last 500 years showed considerable variability among sites, with a range of 1.9 \pm 0.3–74.5 \pm 1.4% and an average of 28.1 \pm 1.2% for the study area (Figure 4d). The mean δ^{13} C signature across sites in the last 500 years ranged from -19.0 ± 0.4 to $-15.3 \pm 0.2\%$ and was on average $-17.3 \pm 0.1\%$ (Figure 5a).





Figure 3. Mean value of cumulative grain size fractions (%) accumulated over the last 500 years at each site.

The contribution of *P. oceanica* detritus to the sediment C_{org} pool ranged from $42 \pm 11\%$ to $74 \pm 4\%$, being on average $60 \pm 10\%$ across sites (Figure 5b). The contribution of other macrophytes and seston was on average (\pm STDEV) 22 $\pm 7\%$, with a range of 17 ± 3 to $32 \pm 13\%$ among sites. Terrestrial material had the lowest contribution to the sediment C_{org} pool, with a mean and range of $18 \pm 5\%$ and $8 \pm 1-27 \pm 2\%$, respectively.

Data on seagrass traits and environmental settings at the study sites are shown in Table S4. The shoot density range was $160 \pm 18-505 \pm 52$ shoots m⁻² across sites, with a mean \pm STDEV of 388 ± 125 shoots m⁻². Leaf biomass ranged between 627 ± 171 and $2,278 \pm 629$ g DW m⁻² and rhizome elongation between 4.8 ± 0.1 and 12.5 ± 0.6 mm yr⁻¹, with mean values of $1,374 \pm 532$ g DW m⁻² and 8.6 ± 2.3 mm yr⁻¹, respectively. Water depth ranged between 3 and 14 m with a mean \pm STDEV of 8 ± 3 m and Secchi depth between 15 and 28 m with a mean of 24 ± 3 m. The concentration of Chl *a* was on average 0.18 ± 0.06 mg m⁻³, varying from 0.12 ± 0.01 to 0.37 ± 0.07 mg m⁻³, and SPM was on average 0.52 ± 0.17 g m⁻³, ranging from 0.38 ± 0.08 to 1.04 ± 0.08 g m⁻³ across sites.

The RDA model explained 70% of the variation in the pattern of C_{org} , N, and $CaCO_3$ accumulation rates, considering all the explanatory variables tested. The explanatory variables that most significantly influenced that variability were the C_{org} :N and C_{org} : C_{inorg} ratios, *P. oceanica* contribution to sediment C_{org} , leaf biomass, rhizome elongation, and SPM (Table 1). C_{org} and N accumulation rates had a strong positive correlation, and both negatively correlated with the CaCO₃ accumulation rate. C_{org} and N accumulation rates were positively correlated with C_{org} :N and C_{org} : C_{inorg} ratios, *P. oceanica* contribution to the sediment C_{org} pool, leaf biomass, and rhizome elongation (Figure 6). The CaCO₃ accumulation rate had a positive correlation with SPM. In total, the greatest part of the variation in the response variables was explained by sediment characteristics (40%), followed by environmental settings (5%) and seagrass traits (2%) (Figure 6).



Journal of Geophysical Research: Biogeosciences



Figure 4. Boxplots of the mean (a) dry bulk density (DBD), (b) C_{org} content, (c) N content, and (d) $CaCO_3$ content over the last 500 years at each site. Each box encompasses the 25% and 75% quantiles, and the horizontal dash indicates the median.

4. Discussion

This study revealed that *P. oceanica* meadows of the South Aegean Sea rank among the most effective carbon and nitrogen sinks reported for seagrasses worldwide. The low variability in habitat geomorphology across the area of study, with all meadows being near the shore (70–450 m from the nearest coast) and away from river mouths or streams (even ephemeral ones), allowed us to understand the influence of the local biogeochemical and environmental context on the differences between organic and inorganic accumulation. Sediment characteristics shaped the pattern of accumulation, contributing the most to explaining the variability among the accumulation of the different elements, followed by the influence of seagrass traits and environmental regime. The large contribution of *P. oceanica* C_{org} into the sedimentary pool reinforces the role of seagrasses as natural blue carbon sinks.

4.1. Corg, N, and CaCO3 Accumulation Rates in P. oceanica Meadows From the South Aegean Sea

Despite the significant increase in research efforts around BC in seagrass ecosystems (Costa et al., 2022), surprisingly we still lack a global robust estimate of the C_{org} accumulation rate in seagrass meadows, apart from early estimations based on a few species (Kennedy et al., 2010; Mcleod et al., 2011) which may render the current value ambiguous. Generally, assessments of C_{org} accumulation rates are undertaken at a local or, at best, at a national level based on very limited data. Comparison with estimates of the long-term C_{org} accumulation rate of seagrass species from other parts of the world shows that our mean value over the last 500 years (33.6 ± 23.6 g $C_{org} m^{-2} yr^{-1}$) lies on the higher end of the range reported for the taxa so far in the Red and Caribbean Seas, as well as the Indian, Pacific, and Southern Oceans (Fu et al., 2023; Lafratta et al., 2020; Miyajima et al., 2022; Rozaimi et al., 2016; Serrano et al., 2018, 2021). The mean C_{org} accumulation rate in this study was lower than the mean value reported for *P. oceanica* meadows in Italy, Spain, and Malta (84 ± 20 g $C_{org} m^{-2} yr^{-1}$) (Serrano,

, 2024, 12, Downloaded

brary.wiley.com/doi/10.1029/2024JG008163 by Thano

, Wiley Online Library on [16/12/2024]. See the Tern



Journal of Geophysical Research: Biogeosciences



Figure 5. Boxplots of the mean (a) δ^{13} C composition and (b) contribution of end-members to sediment C_{org} over the last 500 years at each site. Each box encompasses the 25% and 75% quantiles, and the horizontal dash indicates the median.

Lavery, López-Merino, et al., 2016). However, Serrano, Lavery, López-Merino, et al. (2016) reported data for hotspots of Corg accumulation in shallow meadows where organic-rich escarpments of several meters in thickness were present and thereby, likely provided an overestimate of Corg accumulation in P. oceanica meadows. When compared with data from other studies, our mean C_{org} accumulation rate fell close to the mean rate for the species in Spain (27 ± 16 g C_{org} m⁻² yr⁻¹, Mateo et al., 2019), or even higher than the short-term C_{org} accumulation rate (~100–120 years based on ²¹⁰Pb dating) measured in Spain (26 \pm 15 g C_{org} m⁻² yr⁻¹ (Mazarrasa et al., 2017a), Italy (24 \pm 4 g C_{org} m⁻² yr⁻¹ (Apostolaki et al., 2022), and Cyprus (14 \pm 4 g C_{org} m⁻² yr⁻¹ (Wesselmann et al., 2021)). It is important to note, however, that the aforementioned C_{org} accumulation rates corresponded to very different timeframes; the mean in Serrano, Lavery, López-Merino, et al. (2016) was standardized for a 2-m sediment thickness corresponding to 400 to 3,300 years, while in Mateo et al. (2019) corresponded to 355 to 9,422 years, suggesting that they should be recalculated to 500 years to deliver robust comparisons. In addition, comparisons with short-term accumulation rates should be made with caution, as ¹⁴C-based SAR is usually lower than the SAR derived from excess ²¹⁰Pb, further highlighting the importance of incorporating timescales in BC assessments. Standardizing the timeframe, along with establishing precise chronologies and sedimentation rates are crucial for determining the capacity of seagrass to sequester carbon over meaningful timescales for blue carbon science and related management interventions such as identifying hotspots for blue carbon projects. Given the potential of *P. oceanica* for climate mitigation in the Mediterranean (Pergent-Martini et al., 2021), a coordinated effort to summarize and standardize the available knowledge should be undertaken soon to provide a spatially explicit estimation of the Corg accumulation rate for the region, following similar initiatives in the Nordic region (Krause-Jensen et al., 2022), and inform the implementation of blue carbon crediting accounting frameworks

The mean N accumulation rate of P. oceanica in the South Aegean Sea $(1.00 \pm 0.62 \text{ g N m}^{-2} \text{ yr}^{-1})$ was higher than the mean long-term mean rates reported for the species in Spain (0.59 \pm 0.35 g N m⁻² yr⁻¹, with a range of 0.16-1.03 g N m⁻² yr⁻¹) (Mateo et al., 1997) and the short-term N accumulation rate reported in Italy $(0.6 \pm 0.1 \text{ g N m}^{-2} \text{ yr}^{-1})$ (Apostolaki et al., 2022). It also fell inside the range reported for several other seagrass



Table 1

Summary Table of Redundancy Analysis (RDA) and Statistics Between the Response Variables (C_{org} , N and $CaCO_3$ Accumulation Rates) and the Explanatory Variables

Monte Carlo permutation test for the initial RDA model (999 permutations) with all the explanatory variables tested (PG	C1
scores of grain size fractions, Corg:N and Corg: Cinorg ratios, P. oceanica contribution to sediment Corg, lithology, show	ot
density, leaf biomass, rhizome elongation, water depth, Secchi depth, and SPM)	
	_

	df	Variance	F	p-value		
Model	11	2.10	7.0	0.001		
Residuals	33	0.90				
R^2	0.70					
R ² adjusted	0.60					
Monte Carlo Permutation test with back	kward selection	n in constrained o	ordination (999	permutations)		
	df	F	p-value			
C _{org} :N	1	5.16	0.006			
C _{org} :C _{inorg}	1	31.07	0.001			
P. oceanica contribution	1	4.77	0.008			
Leaf biomass	1	3.59	0.021			
Rhizome elongation	1	4.10	0.017			
Suspended matter (SPM)	1	6.60	0.002			
RDA model with backward selected ex	planatory varia	ables				
Partitioning of variance						
Constrained	0.65					
Unconstrained	0.35					
	RDA1	RDA2	RDA3	PC1	PC2	PC
Importance of components:						
Eigenvalue	1.30	0.44	0.22	0.90	0.09	0.05
Proportion explained	0.43	0.15	0.07	0.30	0.03	0.02
Cumulative proportion explained	0.43	0.58	0.65	0.95	0.98	1.00
Monte Carlo Permutation test for the fir	al RDA model	with the backwa	rd selected expl	anatory variable	s (999 perm	utations
	df	Variance	F	p-value		
Model	6	1.96	11.9	0.001		
Residuals	38	1.04				
R^2	0.65					
\mathbf{R}^2 adjusted	0.60					

species globally (Aoki et al., 2020; Apostolaki et al., 2019; Eyre et al., 2016; Leiva-Dueñas et al., 2023; Martins et al., 2022; Saderne et al., 2018; de los Santos et al., 2023). These comparisons across sites or species should be made with caution, as in the case of comparing the C_{org} accumulation rates, due to the different timescales involved. Nevertheless, the available information on N accumulation rates in seagrass ecosystems remains extremely scarce, despite the recognized role of these ecosystems as coastal filters (McGlathery et al., 2007). Although assessing the pathways of N cycling was outside the scope of this study, research on *Z. marina* indicates that burial dominates the N-removal processes, serving as the main mechanism by which seagrasses achieve their N-filtering function (Aoki et al., 2020). In this sense, *P. oceanica* meadows are expected to significantly contribute to nitrogen filtration along the eastern Mediterranean coasts, where eutrophication events, although usually localized, can occasionally become a significant source of coastal contamination (Simboura et al., 2016). Eutrophication threatens the integrity of seagrasses, possibly compromising the capacity of the latter to support effective actions against the former. The variety of biogeochemical processes involved in fixing and releasing N (Eyre et al., 2016; Prystay et al., 2023; Risgaard-Petersen & Ottosen, 2000), which





Figure 6. Redundancy analysis (RDA) correlation plot (Type II scaling) showing the pattern in the response variables (red vectors) and the backward selected explanatory variables (blue vectors) that mostly drove that variation. CAR, NAR, and CaAR indicate C_{org} , N, and CaCO₃ accumulation rates, respectively. A Venn diagram shows the unique and shared variation (R² adjusted) explained by the most significant variables of each set of variables (sediment characteristics (i.e., C_{org} :N and C_{org} :C_{inorg} ratios and *P. oceanica* contribution to the sediment C_{org} pool), seagrass traits (i.e., leaf biomass and rhizome elongation), and environmental settings (i.e., suspended matter, SPM)).

ultimately determine the magnitude of N burial in the sediments, can vary significantly across biogeochemical and environmental gradients. This variability may disrupt the balance between N input and burial, potentially leading to a decline in N burial due to eutrophication (McGlathery et al., 2007). However, further studies are needed to draw reliable conclusions about how seagrass degradation may impact its capacity to remove N, and thereby, aid the development of N credits and markets.

Compared with other *P. oceanica* meadows, the mean CaCO₃ accumulation rate in this study (405 ± 336 g CaCO₃ m⁻² yr⁻¹) was 7-fold higher than estimates from *P. oceanica* in Spain (54.3 ± 1.9 g CaCO₃ m⁻² yr⁻¹) (Serrano et al., 2012). Considering a mass ratio of 12% C_{inorg} in CaCO₃, the recorded values (7–134 g C_{inorg} m⁻² yr⁻¹) fall within the range reported for other seagrass species (Rozaimi et al., 2016; Saderne et al., 2018). Overall, our mean C_{inorg} accumulation rate (49 ± 40 g C_{inorg} m⁻² yr⁻¹) ranged considerably lower than the short-term global mean (182 ± 94 g C_{inorg} m⁻² yr⁻¹) calculated for seagrasses (Saderne et al., 2019). Again, the different timescales in those studies may add some uncertainty to the comparisons, but, generally, seagrass sediments are shown to support considerable, albeit variable, carbonate stocks (Mazarrasa et al., 2015). This variability is often seen across seagrass species, geomorphologies, and environmental gradients, as these factors affect the calcification by seagrass-associated biota, the trapping of carbonate particles, and the balance between carbonate production and dissolution in seagrass sediments (Saderne et al., 2019). The results obtained highlight the exceptional role of seagrasses in carbonate production, which is comparable to other benthic habitats



including maerl and coralligenous habitats (Canals & Ballesteros, 1997). This reinforces the hypothesis that seagrass meadows contribute to climate change adaptation by generating carbonated sands that help protect against coastal erosion and sea level rise (Duarte, Losada, et al., 2013). On the other hand, the production of $CaCO_3$, which is enhanced by the 3D habitat provided by seagrass meadows for calcifying organisms, results in net CO_2 emissions and thereby, withdrawing on the net CO_2 sequestration by C_{org} burial (Macreadie et al., 2017). However, carbonate production and dissolution processes in seagrass meadows are complex and poorly understood. This, together with uncertainties regarding the net effect of carbonate accumulation on C_{org} preservation and accumulation, the export of alkalinity as a net carbon sink pathway, or the potential allochthonous origin of carbonates buried in the meadows, among others (Howard et al., 2018; van Dam et al., 2021), have precluded the inclusion of carbonates in carbon accounting. Therefore, further research is required to fully understand the implications of carbonates in the carbon cycle both within and beyond seagrass habitats.

4.2. Biotic and Abiotic Drivers of Variability in $\rm C_{org},$ N, and $\rm CaCO_3$ Accumulation Rates Across the South Aegean Sea

The relations identified by statistical analysis between the C_{org} and N accumulation rates and the abiotic and biotic explanatory variables (i.e., sedimentary Corg:N and Corg:Cinorg ratios, P. oceanica contribution to the Corg pool, leaf biomass, rhizome elongation and SPM) suggest that the accumulation of organic matter tended to be higher in inorganic carbon-poor sediments where seagrasses were the main contributors of organic matter, which in turn was the result of healthy meadows that extended in clear waters. The positive associations of C_{org} and N accumulation rates with C_{org} and N contents in the sediment have been widely shown to drive the storage of C_{org} and N, respectively (Kindeberg et al., 2018; Leiva-Dueñas et al., 2023). In several studies of P. oceanica and other seagrass species, Corg and N contents usually show a covariation along the sediment profiles (Apostolaki et al., 2022; Fourqurean, Kendrick, et al., 2012; Fourqurean, Duarte, et al., 2012; Leiva-Dueñas et al., 2023). This is possibly due to the similar timescales in the sequestration and decomposition of these elements that drive the stabilization of organic matter. Underlying sediments in seagrass meadows are typically enriched in C_{org} : N due to seagrass tissue having higher Corg:N than phytoplankton (Mazarrasa et al., 2017b). It is also indicative of large seagrass species, which could be attributed primarily to their elevated biomass and production (Mazarrasa et al., 2018). This is particularly true for *P. oceanica*, which has relatively recalcitrant tissue when compared to other seagrass species (Kaal et al., 2018). This recalcitrance slows down the mineralization, resulting in detritus enriched in Corg

Accordingly, the results of mixing models indicated that the accumulation of organic matter was driven by the input of *P. oceanica*. The C_{org} pool was a mixture of C_{org} deriving mainly from *P. oceanica* and to a lesser extent from allochthonous sources. We acknowledge that deeper layers may have undergone postdepositional diagenesis, potentially modifying the $\delta^{13}C$ signatures, so the results of the models should be interpreted with caution. However, there is still limited knowledge about the potential shifts in $\delta^{13}C$ signals during decomposition, especially over large periods (Kelleway et al., 2022). This has led to similar studies in blue carbon science applying the mixing models either in the first cm of sediment (e.g., 5 cm in Apostolaki et al., 2022) or along the whole sediment profile (e.g., 60 cm in Serrano et al., 2018). All the above suggests that further studies are needed before disentangling the relation between decomposition and $\delta^{13}C$ over meaningful timescales for blue carbon assessments. In any case, potential variability in $\delta^{13}C$ signatures is expected to decrease with sediment depth, following the decrease in the decomposition rate with time (i.e., downcore). In this study, temporal variability of $\delta^{13}C$ along the 500-year profile was small. Given that the meadows were monospecific and extended in the ultraoligotrophic conditions of the eastern Mediterranean, far from any significant terrestrial input, it is reasonable to expect that the contribution of *P. oceanica* detritus in the sedimentary C_{org} pool would be considerable.

Leaf biomass and rhizome elongation had a positive influence on C_{org} and N accumulation rates. The importance of diverse seagrass traits (e.g., shoot density, above- and/or below-ground biomass, and plant elemental content) in driving the storage of C_{org} and, to a lesser extent, N has been shown primarily for *Zostera* spp. (Kim et al., 2022; Lima et al., 2020; Novak et al., 2020; Ricart et al., 2020; Röhr et al., 2016). To our knowledge, this is the first study to examine the effect of seagrass traits on the accumulation rates in *P. oceanica* meadows. This species has been demonstrated to influence the sequestration capacity in multiple species comparisons, owing to its particular traits that drive the variability of C_{org} stocks across all seagrass species, even over large-scale factors such as geomorphology (Kennedy et al., 2022). *P. oceanica* is a long-lived large species, reaching high biomass and production (Duarte et al., 2010). The positive correlation between organic matter accumulation and leaf biomass



and rhizome elongation, which is a proxy for seagrass growth, shows that good seagrass status results in a considerable input of seagrass-derived organic matter in the underlying sediments, thereby directly influencing storage. Contemporarily, P. oceanica has long leaves and forms structurally complex meadows, which modify water flow, increase particle sedimentation, and decrease resuspension, enhancing trapping of allochthonous organic matter (Hendriks et al., 2009) and therefore accumulation in the underlying sediments. Among the traits tested, shoot density did not affect the pattern of accumulation, similar to other studies (Gullström et al., 2018; Kim et al., 2022; Röhr et al., 2016). Shoot density is a generic indicator with a slow response time (Roca et al., 2016). This is particularly true for the long-lived, large seagrass species, like P. oceanica studied here, which can sustain high environmental stress, at least until reaching a tipping point. Any change in the environmental conditions is more likely to be first reflected in a physiological variable (e.g., seagrass nutrient content), or like in this case, a morphological one (i.e., growth), eventually affecting the meadow structure. The sampling area included sites in relatively unaffected conditions, with only a few ones situated close to coastal developed areas. As a result, shoot density is not expected to largely differ to the extent that it would interfere with the capacity of the shoots to trap organic particles. Therefore, the contribution of P. oceanica to Corg input is mainly through its high biomass and growth, rather than the number of shoots per surface area, as it remains relatively constant over time in healthy meadows like those in this study.

The C_{org} and N accumulation rates were negatively influenced by SPM, suggesting that good environmental status fueled the accumulation of organic matter. Sites with high sediment C_{org} and N accumulation rates (e.g., Santorini, Serifos, and Kithnos) held healthy and extensive meadows from the near-surface down to approximately 30 m depth, probably associated with ample light availability that enhanced photosynthesis (Alcoverro et al., 2001) allowing the formation of productive meadows. In contrast, sites with low C_{org} and N accumulation rates (e.g., Anavissos, Sounion, and Metopi) were located near coastal developed areas. Poor environmental conditions, likely due to anthropogenic activities, may have weakened the blue carbon potential of the underlying seagrass sediments due to the accumulation of more labile, allochthonous carbon (Kindeberg et al., 2018; Liu et al., 2023; Mazarrasa et al., 2017b; Novak et al., 2020; Samper-Villarreal et al., 2018). Besides, the negative relation of SPM with C_{org} and N accumulation rates confirmed that one of the reasons for the higher accumulation of organic matter in our study was that the main source of carbon in the underlying sediments was autochthonous. This could remove major concerns around the implementation of conservation and restoration blue carbon projects in southern Greece, owing to the relatively low relevance of allochthonous inputs which could otherwise bring uncertainties into the net role of seagrasses as carbon sinks.

Some of the factors usually driving the sequestration of C_{org} and N, such as the mud content (Dahl et al., 2016; Kindeberg et al., 2018; Leiva-Dueñas et al., 2023) or water depth (Serrano et al., 2014), were not important in our multivariate analysis. The sediment texture in our sites was rather homogeneous, with generally limited variation of each grain size fraction across sites, which may explain the lack of influence of granulometry in the pattern of accumulation. Sediments across sites were composed mainly of fine sands ($47.4 \pm 12.5\%$ on average) with low silt/clay content (18 \pm 15% on average). Mud content has been shown to affect positively the accumulation of organic carbon, mostly in cases of small species where the contribution of terrestrial input is usually significant. However, in large species like P. oceanica, the expected positive relation may be disrupted by the higher contribution of autochthonous input (Serrano, Lavery, Duarte, et al., 2016), as in this case. For example, meadows in Amorgos and Milos, which had ~80% fine sand and silt/clay contents, supported 3-fold lower accumulation rates than the meadows with the highest rates (i.e., Methana, Ios, Santorini, Gouves, Kithnos, Serifos and Santorini 2), despite the latter having similar or even lower fine sand and silt/clay contents (52%-85%). Likewise, the depth range of the studied meadows was relatively small (3-14 m) for the bathymetric range of the species in this region, where lush meadows are found at >15 depths, with a mean depth of 17 m for the species in the Greek Seas (Gerakaris et al., 2021). Since the threshold for a decrease in seagrass production in Greece is at depths >14 m (Litsi-Mizan, 2023), a decrease in accumulation may be expected below that value.

The negative correlations between CaCO₃ accumulation rates and both C_{org} and N accumulation rates, as well as C_{org} : C_{inorg} and C_{org} : N ratios, suggest that meadows with >40% CaCO₃ content are less effective sinks than those growing in CaCO₃-poor sediments. Only Gouves, which had one of the highest CaCO₃ accumulation rates in our data set showed a high C_{org} accumulation rate and a C_{org} : C_{inorg} of 0.89 (above the global mean of 0.74 (Mazarrasa et al., 2015). This suggests a positive feedback between CaCO₃ and C_{org} which possibly resulted in the rapid burial and enhanced preservation of C_{org} at the specific site through high sediment



accretion and burial (Gattuso et al., 1998). This result should be considered with caution given the different timescales in Corg and CaCO3 decay rates and their resulting preservation, and the relative importance of lithogenic carbonates. The variability in the CaCO₃ accumulation rate in our data set was considerable, with a coefficient of variation of 83%, likely originating from different lithogenic settings across sites. The sites with volcanoclastic sediments (i.e., Santorini, Santorini 2, Milos and Methana) had significantly lower mean CaCO₃ content than sites with karstic, siliciclastic, or mixed karstic/siliciclastic sediments (Kruskal-Wallis test and pvalue = 0.02), supporting, on average, 3.4-times lower CaCO₃ accumulation rates (118 \pm 70 g CaCO₃ m⁻² yr⁻¹) than the mean measured across all sites. Volcanic minerals in the sediments could enhance carbonate dissolution, resulting in lower CaCO₃ contents (Vizzini et al., 2019). The remaining sites belonged to the Attico-Cycladic geotectonic zone, which included the sites in the Attica Peninsula, Cyclades archipelago, and northern Crete, where the primary sedimentary rocks were composed of siliciclastic or karstic sediments. Naturally, the seagrass sediments at some of our sites included lithogenic carbonates, yielding high CaCO₃ accumulation rates, as observed in Metopi and Keros. However, the partition between karstic and nonkarstic sediments did not affect the accumulation pattern, as not all sites with high CaCO₃ accumulation rates had karstic sediments. In fact, meadows at Metopi and Keros (i.e., within karstic sediments) and meadows at Agia Marina, Koufonisi, and Gouves (i.e., within nonkarstic sediments) had up to 2-fold higher CaCO₃ accumulation rates than the other counterpart sites surveyed. This implies that the accumulation of carbonates was probably fueled by the calcifying organisms inhabiting the meadow (Frankovich & Fourqurean, 1997). The positive influence of SPM on the CaCO₃ accumulation rate suggests that the accumulation of CaCO₃ was also driven by the deposition of suspended carbonate-bearing particles from the bottom water above the canopy and the subsequent trapping of the latter within the meadow (Gacia et al., 2003).

We acknowledge that the timescale accounted for the sediment variables was different from that of the environmental and seagrass variables. Yet the temporal variability of geochemical variables over the last 500 years was overall low (Table S1), indicating relatively stable accumulation rates. This could be due to the nature of the seagrass-enriched sediments, which are usually refractory, resulting in low decay rates and high preservation, or a result of stable input. Although an increase in C_{org} and N contents was observed at some sites during the last 50 years (e.g., Keros, Koufonisi, Santorini, and Gouves), this trend does not necessarily reflect an increase in the organic matter availability in recent years, as those sites are away from major human activity. Relevant studies in Greek Seas have indicated that human impacts are restricted to specific areas of the country (close to major cities and/or industrial activities) and mainly since the years following post-World War II development (Hatzianestis et al., 2020; Pappa et al., 2018). Therefore, those downcore decreases most probably indicated an initial decay followed by stabilization, suggesting the preservation of C_{org} and N. It must be noted, however, that the chronology of our sediments was based on ¹⁴C dating, which introduces uncertainty when reconstructing the recent past (~50–100 years). To enhance the accuracy of assessing possible changes linked to recent human development in the region, the application of more adequate methods (e.g., ²¹⁰Pb dating), although with their own limitations (Arias-Ortiz et al., 2018), should be considered.

Overall, the analysis of our data suggested that the *P. oceanica* meadows of the South Aegean Sea represent important natural sinks for carbon and nitrogen. Intriguingly, a significant degree of variability was observed within the region, which prompts the need for fine-scale assessments to decipher the intricacies of local conditions influencing carbon and nitrogen sequestration and support effective seagrass management. In our study, the accumulation of organic matter was driven by the contribution of seagrass matter to the sediments, which in turn was a result of healthy meadows extending in clear waters. The significant contribution of *P. oceanica* material, which is normally recalcitrant, in the sedimentary organic matter indicates that these meadows represent permanent pools of elements. This permanence underscores the critical need for conservation efforts to safeguard these coastal ecosystems and prevent the potential loss of centennial stored carbon and nitrogen following seagrass degradation, which could contribute to accelerate climate change and exacerbate eutrophication.

Data Availability Statement

The data supporting this study are available and hosted in the SEANOE database: https://www.seanoe.org/data/00885/99669 (Apostolaki et al., 2024).



Acknowledgments

This work was supported by ARC DECRA (DE170101524), PID2020-117639GB-100 funded by MCIN/AEI, DRESSAGE (MIS5045792) through the Operational Program "Competitiveness, Entrepreneurship and Innovation" (EPAnEK 2014–2020, NSRF), and MEDCHANGE funded by the Agencia Española de Investigación. The publication of the article in OA mode was financially supported by HEAL-Link. We thank the reviewers for their constructive comments.

References

- Aitchison, J. (1986). The statistical analysis of compositional data. In *Monographs on Statistics and Applied Probability*. Chapman and Hall Ltd. Alcoverro, T., Cerbian, E., & Ballesteros, E. (2001). The photosynthetic capacity of the seagrass *Posidonia oceanica*: Influence of nitrogen and light. *Journal of Experimental Marine Biology and Ecology*, 261(1), 107–120. https://doi.org/10.1016/s0022-0981(01)00267-2
- Aoki, L. R., McGlathery, K. J., & Oreska, M. P. J. (2020). Seagrass restoration reestablishes the coastal nitrogen filter through enhanced burial. Linnology & Oceanography, 65(1), 1–12. https://doi.org/10.1002/ino.11241
- Apostolaki, E. T., Caviglia, L., Santinelli, V., Cundy, A. B., Tramati, C. D., Mazzola, A., & Vizzini, S. (2022). The importance of Dead seagrass (*Posidonia oceanica*) matte as a biogeochemical sink. *Frontiers in Marine Science*, 9, 861998. https://doi.org/10.3389/fmars.2022.861998
- Apostolaki, E. T., Vizzini, S., Santinelli, V., Kaberi, H., Andolina, C., & Papathanassiou, E. (2019). Exotic Halophila stipulacea is an introduced carbon sink for the Eastern Mediterranean Sea. Scientific Reports, 9(1), 9643. https://doi.org/10.1038/s41598-019-45046-w
- Apostolaki Eugenia, T., Lavery Paul, S., Victoria, L.-M., Eduard, S., Karina, I., Vasilis, G., et al. (2024). Carbon and nitrogen stocks and accumulation rates in seagrass meadows (*Posidonia oceanica*) of the Eastern Mediterranean Sea [Dataset]. SEANOE. https://doi.org/10.17882/ 99669
- Arias-Ortiz, A., Masqué, P., Garcia-Orellana, J., Serrano, O., Mazarrasa, I., Marbà, N., et al. (2018). Reviews and syntheses: ²¹⁰Pb-derived sediment and carbon accumulation rates in vegetated coastal ecosystems – Setting the record straight. *Biogeosciences*, 15(22), 6791–6818. https://doi.org/10.5194/bg-15-6791-2018-supplement
- Barrón, C., Apostolaki, E. T., & Duarte, C. M. (2014). Dissolved organic carbon fluxes by seagrass meadows and macroalgal beds. Frontiers in Marine Science, 1, 42. https://doi.org/10.3389/fmars.2014.00042/abstract
- Blaauw, M., & Christen, J. A. (2011). Flexible paleoclimate age-depth models using an autoregressive gamma process. Bayesian Anal, 6(3), 457–474. https://doi.org/10.1214/11-BA618
- Canals, M., & Ballesteros, E. (1997). Production of carbonate particles by phytobenthic communities on the Mallorca-Menorca shelf, northwestern Mediterranean Sea. Deep Sea Research Part II: Topical Studies in Oceanography, 44(3–4), 611–629. https://doi.org/10.1016/s0967-0645(96)00095-1
- Costa, M. D., De, P., & Macreadie, P. I. (2022). The evolution of blue carbon science. Wetlands, 42(8), 109. https://doi.org/10.1007/s13157-022-01628-5
- Dahl, M., Deyanova, D., Gütschow, S., Asplund, M. E., Lyimo, L. D., Karamfilov, V., et al. (2016). Sediment properties as important predictors of carbon storage in *Zostera marina* meadows: A comparison of four European areas. *PLoS One*, 11(12), e0167493. https://doi.org/10.1371/ journal.pone.0167493
- de los Santos, C. B., Egea, L. G., Martins, M., Santos, R., Masqué, P., Peralta, G., et al. (2023). Sedimentary organic carbon and nitrogen sequestration across a vertical gradient on a temperate Wetland seascape including salt marshes, seagrass meadows and rhizophytic macroalgae beds. *Ecosystems*, 26(4), 826–842. https://doi.org/10.1007/s10021-022-00801-5
- Duarte, C. M., Kennedy, H., Marbà, N., & Hendriks, I. (2013). Assessing the capacity of seagrass meadows for carbon burial: Current limitations and future strategies. Ocean and Coastal Management, 83, 32–38. https://doi.org/10.1016/j.ocecoaman.2011.09.001
- Duarte, C. M., Losada, I. J., Hendriks, I. E., Mazarrasa, I., & Marbà, N. (2013). The role of coastal plant communities for climate change mitigation and adaptation. *Nature Climate Change*, 3(11), 961–968. https://doi.org/10.1038/nclimate1970
- Duarte, C. M., Marba, N., Gacia, E., Fourqurean, J. W., Beggins, J., Barrón, C., & Apostolaki, E. T. (2010). Seagrass community metabolism: Assessing the carbon sink capacity of seagrass meadows. *Global Biogeochemical Cycles*, 24(4), GB4032. https://doi.org/10.1029/ 2010GB003793
- Eyre, B. D., Maher, D. T., & Sanders, C. (2016). The contribution of denitrification and burial to the nitrogen budgets of three geomorphically distinct Australian estuaries: Importance of seagrass habitats. *Limnology & Oceanography*, 61(3), 1144–1156. https://doi.org/10.1002/lno. 10280
- Fourqurean, J. W., Duarte, C. M., Kennedy, H., Marbà, N., Holmer, M., Mateo, M.-Á., et al. (2012). Seagrass ecosystems as a globally significant carbon stock. *Nature Geoscience*, 5(6), 505–509. https://doi.org/10.1038/ngeo1477
- Fourqurean, J. W., Kendrick, G. A., Collins, L. S., Chambers, R. M., & Vanderklift, M. A. (2012). Carbon, nitrogen and phosphorus storage in subtropical seagrass meadows: Examples from Florida Bay and Shark Bay. *Marine and Freshwater Research*, 63(11), 967. https://doi.org/10. 1071/mf12101
- Frankovich, T. A., & Fourqurean, J. W. (1997). Seagrass epiphyte loads along a nutrient availability gradient, Florida Bay, USA. Marine Ecology Progress Series, 159, 37–50. https://doi.org/10.3354/meps159037
- Fu, C., Frappi, S., Havlik, M. N., Howe, W., Harris, S. D., Laiolo, E., et al. (2023). Substantial blue carbon sequestration in the world's largest seagrass meadow. *Communications Earth & Environment*, 4(1), 474. https://doi.org/10.1038/s43247-023-01154-0
- Gacia, E., Duarte, C. M., Marba, N., Terrados, J., Kennedy, H., Fortes, M. D., & Tri, N. H. (2003). Sediment deposition and production in SE-Asia seagrass meadows. *Estuarine, Coastal and Shelf Science*, 56(5–6), 909–919. https://doi.org/10.1016/s0272-7714(02)00286-x
- Gacia, E., Duarte, C. M., & Middelburg, J. J. (2002). Carbon and nutrient deposition in a Mediterranean seagrass (*Posidonia oceanica*) meadow. *Limnology & Oceanography*, 47(1), 23–32. https://doi.org/10.4319/lo.2002.47.1.0023
- Garcias-Bonet, N., Fusi, M., Ali, M., Shaw, D. R., Saikaly, P. E., Daffonchio, D., & Duarte, C. M. (2018). High denitrification and anaerobic ammonium oxidation contributes to net nitrogen loss in a seagrass ecosystem in the central Red Sea. *Biogeosciences*, 15(23), 7333–7346. https://doi.org/10.5194/bg-15-7333-2018
- Gattuso, J. P., Frankignoulle, M., & Wollast, R. (1998). Carbon and carbonate metabolism in coastal aquatic ecosystems. Annual Review of Ecology and Systematics, 29(1), 405–434. https://doi.org/10.1146/annurev.ecolsys.29.1.405
- Gerakaris, V., Papathanasiou, V., Salomidi, M., Issaris, Y., & Panayotidis, P. (2021). Spatial patterns of *Posidonia oceanica* structural and functional features in the Eastern Mediterranean (Aegean and E. Ionian Seas) in relation to large-scale environmental factors. *Marine Envi*ronmental Research, 165, 105222. https://doi.org/10.1016/j.marenvres.2020.105222
- Gullström, M., Lyimo, L. D., Dahl, M., Samuelsson, G. S., Eggertsen, M., Anderberg, E., et al. (2018). Blue carbon storage in tropical seagrass meadows relates to carbonate stock dynamics, plant–sediment processes, and landscape context: Insights from the western Indian ocean. *Ecosystems*, 21(3), 551–566. https://doi.org/10.1007/s10021-017-0170-8
- Hatzianestis, I., Parinos, C., Bouloubassi, I., & Gogou, A. (2020). Polycyclic aromatic hydrocarbons in surface sediments of the Aegean Sea (Eastern Mediterranean Sea). Marine Pollution Bulletin, 153, 111030. https://doi.org/10.1016/j.marpolbul.2020.111030
- Hendriks, I. E., Bouma, T. J., Morris, E. P., & Duarte, C. M. (2009). Effects of seagrasses and algae of the Caulerpa family on hydrodynamics and particle-trapping rates. *Marine Biology*, 157(3), 473–481. https://doi.org/10.1007/s00227-009-1333-8
- Howard, J. L., Creed, J. C., Aguiar, M. V., & Fourqurean, J. W. (2018). CO₂ released by carbonate sediment production in some coastal areas may offset the benefits of seagrass "Blue Carbon" storage. *Limnology & Oceanography*, 63(1), 160–172. https://doi.org/10.1002/lno.10621

2169896

, 2024, 12, Downloaded

com/doi/10.1029/2024JG008163 by Thance

, Wiley

Unline

Library on

- Institute of Geology and Mineral Exploration (IGME), Division of General Geology and Economic Geology. (1983). Geological map of Greece, scale 1:500000. Redaction: Bornovas J., Rondogianni-Tsiambaou Th (2nd ed.). Institute of Geology and Mineral Exploration (IGME).
- Kaal, J., Serrano, O., Cortizas, A. M., Baldock, J. A., & Lavery, P. S. (2019). Millennial-scale changes in the molecular composition of *Posidonia australis* seagrass deposits: Implications for Blue Carbon sequestration. *Organic Geochemistry*, 137, 103898. https://doi.org/10.1016/j.orggeochem.2019.07.007
- Kaal, J., Serrano, O., del Río, J. C., & Rencoret, J. (2018). Radically different lignin composition in Posidonia species may link to differences in organic carbon sequestration capacity. Organic Geochemistry, 124, 247–256. https://doi.org/10.1016/j.orggeochem.2018.07.017
- Kelleway, J. J., Trevathan-Tackett, S. M., Baldock, J., & Critchley, L. P. (2022). Plant litter composition and stable isotope signatures vary during decomposition in blue carbon ecosystems. *Biogeochemistry*, 158(2), 147–165. https://doi.org/10.1007/s10533-022-00890-3
- Kennedy, H., Beggins, J., Duarte, C. M., Fourqurean, J. W., Holmer, M., Marba, N., & Middelburg, J. J. (2010). Seagrass sediments as a global carbon sink: Isotopic constraints. *Global Biogeochemical Cycles*, 24(4), GB4026. https://doi.org/10.1029/2010gb003848
- Kennedy, H., Pagès, J. F., Lagomasino, D., Arias-Ortiz, A., Colarusso, P., Fourqurean, J. W., et al. (2022). Species traits and geomorphic setting as drivers of global soil carbon stocks in seagrass meadows. *Global Biogeochemical Cycles*, 36(10), e2022GB007481. https://doi.org/10.1029/ 2022gb007481
- Kennedy, P., Kennedy, H., & Papadimitriou, S. (2005). The effect of acidification on the determination of organic carbon, total nitrogen and their stable isotopic composition in algae and marine sediment. *Rapid Communications in Mass Spectrometry*, 19(8), 1063–1068. https://doi.org/10. 1002/rcm.1889
- Kim, S. H., Suonan, Z., Qin, L.-Z., Kim, H., Park, J.-I., Kim, Y. K., et al. (2022). Variability in blue carbon storage related to biogeochemical factors in seagrass meadows off the coast of the Korean peninsula. *Science of the Total Environment*, 813, 152680. https://doi.org/10.1016/j. scitotenv.2021.152680
- Kindeberg, T., Ørberg, S. B., Röhr, M. E., Holmer, M., & Krause-Jensen, D. (2018). Sediment stocks of carbon, nitrogen, and phosphorus in Danish eelgrass meadows. *Frontiers in Marine Science*, 5, 474. https://doi.org/10.3389/fmars.2018.00474
- Krause-Jensen, D., Gundersen, H., Björk, M., Gullström, M., Dahl, M., Asplund, M. E., et al. (2022). Nordic blue carbon ecosystems: Status and outlook. Frontiers in Marine Science, 9, 847544. https://doi.org/10.3389/fmars.2022.847544
- Lafratta, A., Serrano, O., Masqué, P., Mateo, M. A., Fernandes, M., Gaylard, S., & Lavery, P. S. (2020). Challenges to select suitable habitats and demonstrate 'additionality' in blue carbon projects: A seagrass case study. Ocean and Coastal Management, 197, 105295. https://doi.org/10. 1016/j.ocecoaman.2020.105295
- Lavery, P. S., Mateo, M.-Á., Serrano, O., & Rozaimi, M. (2013). Variability in the carbon storage of seagrass habitats and its implications for global estimates of blue carbon ecosystem service. PLoS One, 8(9), e73748. https://doi.org/10.1371/journal.pone.0073748.t008
- Leiva-Dueñas, C., Graversen, A. E. L., Banta, G. T., Holmer, M., Masque, P., Stæhr, P. A. U., & Krause-Jensen, D. (2023). Capturing of organic carbon and nitrogen in eelgrass sediments of southern Scandinavia. *Limnology & Oceanography*, 68(3), 631–648. https://doi.org/10.1002/lno. 12299
- Lima, M., Ward, R. D., & Joyce, C. B. (2020). Environmental drivers of sediment carbon storage in temperate seagrass meadows. *Hydrobiologia*, 847(7), 1773–1792. https://doi.org/10.1007/s10750-019-04153-5
- Litsi-Mizan, V. (2023). Understanding the function and vulnerability of seagrass (Posidonia oceanica) meadows along the Greek Seas to environmental change. PhD Thesis (p. 198). University of Crete.
- Liu, S., Luo, H., Jiang, Z., Ren, Y., Zhang, X., Wu, Y., et al. (2023). Nutrient loading weakens seagrass blue carbon potential by stimulating seagrass detritus carbon emission. *Ecological Indicators*, 157, 111251. https://doi.org/10.1016/j.ecolind.2023.111251
- Macreadie, P. I., Costa, M. D. P., Atwood, T. B., Friess, D. A., Kelleway, J. J., Kennedy, H., et al. (2021). Blue carbon as a natural climate solution. *Nature Reviews Earth and Environment*, 2(12), 826–839. https://doi.org/10.1038/s43017-021-00224-1
- Macreadie, P. I., Serrano, O., Maher, D. T., Duarte, C. M., & Beardall, J. (2017). Addressing calcium carbonate cycling in blue carbon accounting. Limnology & Oceanography, 31, 513. https://doi.org/10.1016/j.earscirev.2016.03.003
- Martins, M., de los Santos, C. B., Masqué, P., Carrasco, A. R., Veiga-Pires, C., & Santos, R. (2022). Carbon and nitrogen stocks and burial rates in intertidal vegetated habitats of a Mesotidal coastal Lagoon. *Ecosystems*, 25(2), 372–386. https://doi.org/10.1007/s10021-021-00660-6
- Mateo, M. A., Diaz-Almela, E., Piñeiro-Juncal, N., Leiva-Dueñas, C., Giralt-Romeu, S., & Marco-Méndez, C. (2019). Carbon stocks and fluxes associated to Andalusian seagrasses: Variability and effects of meadow degradation and recolonization (p. 145). LIFE Blue Natura project. Deliverable C1.2.
- Mateo, M. A., Romero, J., Perez, M., Littler, M. M., & Littler, D. S. (1997). Dynamics of millenary organic deposits resulting from the growth of the Mediterranean seagrass Posidonia oceanica. *Estuarine, Coastal and Shelf Science*, 44(1), 103–110. https://doi.org/10.1006/ecss.1996.0116
- Mazarrasa, I., Lavery, P., Duarte, C. M., Lafratta, A., Lovelock, C. E., Macreadie, P. I., et al. (2021). Factors determining seagrass blue carbon across bioregions and geomorphologies. *Global Biogeochemical Cycles*, 35(6), e2021GB006935. https://doi.org/10.1029/2021gb006935
- Mazarrasa, I., Marbà, N., Garcia-Orellana, J., Masqué, P., Arias-Ortiz, A., & Duarte, C. M. (2017a). Dynamics of carbon sources supporting burial in seagrass sediments under increasing anthropogenic pressure. *Limnology & Oceanography*, 62(4), 1451–1465. https://doi.org/10.1002/lno. 10509
- Mazarrasa, I., Marbà, N., Garcia-Orellana, J., Masqué, P., Arias-Ortiz, A., & Duarte, C. M. (2017b). Effect of environmental factors (wave exposure and depth) and anthropogenic pressure in the C sink capacity of Posidonia oceanica meadows. *Limnology & Oceanography*, 62(4), 1436–1450. https://doi.org/10.1002/lno.10510
- Mazarrasa, I., Marba, N., Lovelock, C. E., Serrano, O., Lavery, P. S., Fourqurean, J. W., et al. (2015). Seagrass meadows as a globally significant carbonate reservoir. *Biogeosciences Discussions*, 12(5), 4107–4138. https://doi.org/10.5194/bgd-12-4993-2015
- Mazarrasa, I., Samper-Villarreal, J., Serrano, O., Lavery, P. S., Lovelock, C. E., Marbà, N., et al. (2018). Habitat characteristics provide insights of carbon storage in seagrass meadows. *Marine Pollution Bulletin*, 134, 106–117. https://doi.org/10.1016/j.marpolbul.2018.01.059
- McGlathery, K. J., Reynolds, L. K., Cole, L. W., Orth, R. J., Marion, S. R., & Schwarzschild, A. (2012). Recovery trajectories during state change from bare sediment to eelgrass dominance. *Marine Ecology Progress Series*, 448, 209–221. https://doi.org/10.3354/meps09574
- McGlathery, K. J., Sundback, K., & Anderson, I. C. (2007). Eutrophication in shallow coastal bays and lagoons: The role of plants in the coastal filter. *Marine Ecology Progress Series*, 348, 1–18. https://doi.org/10.3354/meps07132
- Mcleod, E., Chmura, G. L., Bouillon, S., Salm, R., Björk, M., Duarte, C. M., et al. (2011). A blueprint for blue carbon: Toward an improved understanding of the role of vegetated coastal habitats in sequestering CO₂. Frontiers in Ecology and the Environment, 9(10), 552–560. https:// doi.org/10.1016/j.quascirev.2010.03.003
- Miyajima, T., Hamaguchi, M., & Hori, M. (2022). Evaluation of the baseline carbon sequestration rates of Indo-Pacific temperate and tropical seagrass meadow sediments. *Ecological Research*, 37(1), 9–20. https://doi.org/10.1111/1440-1703.12263



- Monnier, B., Pergent, G., Mateo, M. Á., Carbonell, R., Clabaut, P., & Pergent-Martini, C. (2021). Sizing the carbon sink associated with *Posidonia* oceanica seagrass meadows using very high-resolution seismic reflection imaging. *Marine Environmental Research*, 170, 105415. https://doi. org/10.1016/j.marenvres.2021.105415
- Monnier, B., Pergent, G., Mateo, M. Á., Clabaut, P., & Pergent-Martini, C. (2022). Quantification of blue carbon stocks associated with *Posidonia* oceanica seagrass meadows in Corsica (NW Mediterranean). Science of the Total Environment, 838(Pt 1), 155864. https://doi.org/10.1016/j. scitotenv.2022.155864
- Novak, A. B., Pelletier, M. C., Colarusso, P., Simpson, J., Gutierrez, M. N., Arias-Ortiz, A., et al. (2020). Factors influencing carbon stocks and accumulation rates in eelgrass meadows across New England, USA. *Estuaries and Coasts*, 43(8), 2076–2091. https://doi.org/10.1007/s12237-020-00754-9
- Oksanen, J., Simpson, G., Blanchet, F., Kindt, R., Legendre, P., Minchin, P., et al. (2022). vegan: Community Ecology Package. R package version 2.6-4. Retrieved from https://CRAN.R-project.org/package=vegan
- Panayotidis, P., Papathanasiou, V., Gerakaris, V., Fakiris, E., Orfanidis, S., Papatheodorou, G., et al. (2022). Seagrass meadows in the Greek seas: Presence, abundance and spatial distribution. *Botanica Marina*, 65(4), 289–299. https://doi.org/10.1515/bot-2022-0011
- Pappa, F. K., Tsabaris, C., Patiris, D. L., Androulakaki, E. G., Eleftheriou, G., Betsou, C., et al. (2018). Historical trends and assessment of radionuclides and heavy metals in sediments near an abandoned mine, Lavrio, Greece. *Environmental Science and Pollution Research*, 25(30), 30084–30100. https://doi.org/10.1007/s11356-018-2984-0
- Parnell, A. C., Inger, R., Bearhop, S., & Jackson, A. L. (2010). Source partitioning using stable isotopes: Coping with too much variation. PLoS One, 5(3), 5. https://doi.org/10.1371/journal.pone.0009672
- Parnell, A. C., Phillips, D. L., Bearhop, S., Semmens, B. X., Ward, E. J., Moore, J. W., et al. (2013). Bayesian stable isotope mixing models. *Environmetrics*, 24(6), 387–399. https://doi.org/10.1002/env.2221
- Peirano, A. (2002). Lepidochronology and internodal length methods for studying *Posidonia oceanica* growth: Are they compatible? *Aquatic Botany*, 74(2), 175–180. https://doi.org/10.1016/s0304-3770(02)00078-5
- Pergent-Martini, C., & Pergent, G. (1994). Lepidochronological analysis in the Mediterranean seagrass *Posidonia oceanica*: State of the art and future developments. *Oceanologica Acta*, 17, 673–681.
- Pergent-Martini, C., Pergent, G., Monnier, B., Boudouresque, C.-F., Mori, C., & Valette-Sansevin, A. (2021). Contribution of *Posidonia oceanica* meadows in the context of climate change mitigation in the Mediterranean Sea. *Marine Environmental Research*, 165, 105236. https://doi.org/ 10.1016/j.marenvres.2020.105236
- Prystay, T. S., Sipler, R. E., Foroutani, M. B., & Bris, A. L. (2023). The role of boreal seagrass meadows in the coastal filter. Journal of Geophysical Research: Biogeosciences, 128(12). https://doi.org/10.1029/2023jg007537
- R Core Team. (2021). R: A language and environment for statistical computing. Retrieved from https://www.R-project.org/
- Reimer, P. J., & McCormac, F. G. (2002). Marine radiocarbon reservoir corrections for the Mediterranean and Aegean seas. *Radiocarbon*, 44(1), 159–166. https://doi.org/10.1017/s0033822200064766
- Reynolds, L. K., Waycott, M., McGlathery, K. J., & Orth, R. J. (2016). Ecosystem services returned through seagrass restoration. *Restoration Ecology*, 24(5), 583–588. https://doi.org/10.3354/meps208065
- Ricart, A. M., York, P. H., Bryant, C. V., Rasheed, M. A., Ierodiaconou, D., & Macreadie, P. I. (2020). High variability of Blue Carbon storage in seagrass meadows at the estuary scale. *Scientific Reports*, 10(1), 5865. https://doi.org/10.1038/s41598-020-62639-y
- Risgaard-Petersen, N., & Ottosen, L. D. M. (2000). Nitrogen cycling in two temperate Zostera marina beds: Seasonal variation. Marine Ecology Progress Series, 198, 93–107. https://doi.org/10.3354/meps198093
- Roca, G., Alcoverro, T., Krause-Jensen, D., Balsby, T. J. S., van Katwijk, M. M., Marbà, N., et al. (2016). Response of seagrass indicators to shifts in environmental stressors: Aglobal review and management synthesis. *Ecological Indicators*, 63, 310–323. https://doi.org/10.1016/j.ecolind. 2015.12.007
- Röhr, M. E., Boström, C., Canal-Vergés, P., & Holmer, M. (2016). Blue carbon stocks in Baltic Sea eelgrass Zostera marina meadows. *Biogeosciences*, 13(22), 6139–6153. https://doi.org/10.5194/bg-13-6139-2016
- Röhr, M. E., Holmer, M., Baum, J. K., Björk, M., Boyer, K., Chin, D., et al. (2018). Blue carbon storage capacity of temperate eelgrass (Zostera marina) meadows. *Global Biogeochemical Cycles*, 32(10), 1457–1475. https://doi.org/10.1029/2018gb005941
- Romero, J., Perez, M., Mateo, M. A., & Sala, E. (1994). The belowground organs of the Mediterranean seagrass Posidonia oceanica as a biogeochemical sink. Aquatic Botany, 47(1), 13–19. https://doi.org/10.1016/0304-3770(94)90044-2
- Rozaimi, M., Lavery, P. S., Serrano, O., & Kyrwood, D. (2016). Long-term carbon storage and its recent loss in an estuarine *Posidonia australis* meadow (Albany, Western Australia). *Estuarine, Coastal and Shelf Science*, 171(C), 58–65. https://doi.org/10.1016/j.ecss.2016.01.001
- Saderne, V., Cusack, M., Almahasheer, H., Serrano, O., Masqué, P., Arias-Ortiz, A., et al. (2018). Accumulation of carbonates contributes to coastal vegetated ecosystems keeping pace with sea level rise in an arid region (Arabian peninsula). *Journal of Geophysical Research: Biogeosciences*, 123(5), 1498–1510. https://doi.org/10.1029/2017jg004288
- Saderne, V., Geraldi, N. R., Macreadie, P. I., Maher, D. T., Middelburg, J. J., Serrano, O., et al. (2019). Role of carbonate burial in Blue Carbon budgets. *Nature Communications*, 10(1), 1106. https://doi.org/10.1038/s41467-019-08842-6
- Samper-Villarreal, J., Lovelock, C. E., Saunders, M. I., Roelfsema, C., & Mumby, P. J. (2016). Organic carbon in seagrass sediments is influenced by seagrass canopy complexity, turbidity, wave height, and water depth. *Limnology & Oceanography*, 61(3), 938–952. https://doi.org/10.1002/ lno.10262
- Samper-Villarreal, J., Mumby, P. J., Saunders, M. I., Barry, L. A., Zawadzki, A., Heijnis, H., et al. (2018). Vertical accretion and carbon burial rates in subtropical seagrass meadows increased following anthropogenic pressure from European colonisation. *Estuarine, Coastal and Shelf Science*, 202, 40–53. https://doi.org/10.1016/j.ecss.2017.12.006
- Serrano, O., Almahasheer, H., Duarte, C. M., & Irigoien, X. (2018). Carbon stocks and accumulation rates in Red Sea seagrass meadows. Scientific Reports, 8(1), 1–13. https://doi.org/10.1038/s41598-018-33182-8
- Serrano, O., Gómez-López, D. I., Sánchez-Valencia, L., Acosta-Chaparro, A., Navas-Camacho, R., González-Corredor, J., et al. (2021). Seagrass blue carbon stocks and sequestration rates in the Colombian Caribbean. *Scientific Reports*, 11(1), 11067. https://doi.org/10.1038/s41598-021-90544-5
- Serrano, O., Lavery, P. S., Duarte, C. M., Kendrick, G. A., Calafat, A., York, P. H., et al. (2016). Can mud (silt and clay) concentration be used to predict soil organic carbon content within seagrass ecosystems? *Biogeosciences*, 13(17), 4915–4926. https://doi.org/10.4225/75/ 56c55ab91d417
- Serrano, O., Lavery, P. S., López-Merino, L., Ballesteros, E., & Mateo, M. A. (2016). Location and associated carbon storage of erosional escarpments of seagrass Posidonia Mats. Frontiers in Marine Science, 3, 42. https://doi.org/10.3389/fmars.2016.00042
- Serrano, O., Lavery, P. S., Rozaimi, M., & Mateo, M. Á. (2014). Influence of water depth on the carbon sequestration capacity of seagrasses. *Global Biogeochemical Cycles*, 28(9), 950–961. https://doi.org/10.1002/2014gb004872



- Serrano, O., Mateo, M. A., Renom, P., & Julià, R. (2012). Characterization of soils beneath a Posidonia oceanica meadow. Geoderma, 185–186, 26–36. https://doi.org/10.1016/j.geoderma.2012.03.020
- Simboura, N., Pavlidou, A., Bald, J., Tsapakis, M., Pagou, K., Zeri, C., et al. (2016). Response of ecological indices to nutrient and chemical contaminant stress factors in Eastern Mediterranean coastal waters. *Ecological Indicators*, 70, 89–105. https://doi.org/10.1016/j.ecolind.2016. 05.018
- Smith, S. V. (1981). Marine macrophytes as a global carbon sink. *Science*, 211(4484), 838–840. https://doi.org/10.1126/science.211.4484.838 Spivak, A. C., Sanderman, J., Bowen, J. L., Canuel, E. A., & Hopkinson, C. S. (2019). Global-change controls on soil-carbon accumulation and
- loss in coastal vegetated ecosystems. *Nature Geoscience*, *12*(9), 685–692. https://doi.org/10.1038/s41561-019-0435-2 Stuiver, M., & Polach, H. A. (1977). Discussion reporting of ¹⁴C data. *Radiocarbon*, *19*(3), 355–363. https://doi.org/10.1017/
- S0033822200003672 Stuiver, M., & Reimer, P. J. (1993). Extended ¹⁴C data base and revised CALIB 3.0 14C age calibration Program. *Radiocarbon*, *35*(1), 215–230.
- Stuiver, M., & Reimer, P. J. (1993). Extended ¹C data base and revised CALIB 3.0 14C age calibration Program. *Radiocarbon*, 35(1), 215–230. https://doi.org/10.1017/s0033822200013904
- Trevathan-Tackett, S. M., Jeffries, T. C., Macreadie, P. I., Manojlovic, B., & Ralph, P. (2019). Long-term decomposition captures key steps in microbial breakdown of seagrass litter. Science of the Total Environment, 705, 135806. https://doi.org/10.1016/j.scitotenv.2019.135806
- Van Dam, B. R., Zeller, M. A., Lopes, C., Smyth, A. R., Böttcher, M. E., Osburn, C. L., et al. (2021). Calcification-driven CO₂ emissions exceed "Blue Carbon" sequestration in a carbonate seagrass meadow. *Science Advances*, 7(51), eabj1372. https://doi.org/10.1126/sciadv.abj1372
- Vehtari, A., Gelman, A., Simpson, D., Carpenter, B., & Bürkner, P.-C. (2019). Rank-normalization, folding, and localization: An improved R-hat for assessing convergence of MCMC. arXiv preprint arXiv:1903.08008.
- Vizzini, S., Apostolaki, E. T., Ricevuto, E., Polymenakou, P., & Mazzola, A. (2019). Plant and sediment properties in seagrass meadows from two Mediterranean CO2 vents: Implications for carbon storage capacity of acidified oceans. *Marine Environmental Research*, 146, 101–108. https://doi.org/10.1016/j.marenvres.2019.03.001
- Wesselmann, M., Geraldi, N. R., Duarte, C. M., Garcia-Orellana, J., Díaz-Rúa, R., Arias-Ortiz, A., et al. (2021). Seagrass (Halophila stipulacea) invasion enhances carbon sequestration in the Mediterranean Sea. *Global Change Biology*, 27(11), 2592–2607. https://doi.org/10.1111/gcb. 15589

References From the Supporting Information

- Apostolopoulou, M. V., Monteyne, E., Krikonis, K., Pavlopoulos, K., Roose, P., & Dehairs, F. (2015). n-Alkanes and stable C, N isotopic compositions as identifiers of organic matter sources in *Posidonia oceanica* meadows of Alexandroupolis Gulf, NE Greece. *Marine Pollution Bulletin*, 99(1–2), 346–355. https://doi.org/10.1016/j.marpolbul.2015.07.033
- Deudero, S., Box, A., Alós, J., Arroyo, N. L., & Marbà, N. (2011). Functional changes due to invasive species: Food web shifts at shallow Posidonia oceanica seagrass beds colonized by the alien macroalga Caulerpa racemosa. Estuarine, Coastal and Shelf Science, 93(2), 106–116. https://doi.org/10.1016/j.ecss.2011.03.017
- Holmer, M., Duarte, C. M., Boschker, H. T. S., & Barrón, C. (2004). Carbon cycling and bacterial carbon sources in pristine and impacted Mediterranean seagrass sediments. Aquatic Microbial Ecology, 36(3), 227–237. https://doi.org/10.3354/ame036227
- Holmer, M., Marba, N., Diaz-Almela, E., Duarte, C. M., Tsapakis, M., & Danovaro, R. (2007). Sedimentation of organic matter from fish farms in oligotrophic Mediterranean assessed through bulk and stable isotope (δ13C and δ15N) analyses. Aquaculture, 262(2–4), 268–280. https://doi. org/10.1016/j.aquaculture.2006.09.033
- Vizzini, S., Sara, G., Michener, R. H., & Mazzola, A. J. A. O. (2002). The role and contribution of the seagrass *Posidonia oceanica* (L.) Delile organic matter for secondary consumers as revealed by carbon and nitrogen stable isotope analysis. *Acta Oecologica*, 23(4), 277–285. https:// doi.org/10.1016/S1146-609X(02)01156-6
- Vizzini, S., Colombo, F., Costa, V., & Mazzola, A. (2012). Contribution of planktonic and benthic food sources to the diet of the reef-forming vermetid gastropod Dendropoma petraeum in the western Mediterranean. Estuarine. *Coastal and Shelf Science*, 96, 262–267. https://doi.org/10. 1016/j.ecss.2011.11.021